Author’s response to reviews

Title: Knockdown of TRPM7 prevents tumor growth, migration, and invasion through the Src, Akt, and JNK pathway in bladder cancer

Authors:

Eun Hye Lee (eun90hye@gmail.com)
So Young Chun (soyachun99@naver.com)
Bomi Kim (bomi9224@naver.com)
Bo Hyun Yoon (bobo1904@naver.com)
Jun Nyung Lee (ljinlover@gmail.com)
Bum Soo Kim (dock97@hanmail.net)
Eun Sang Yoo (uroyoo@knu.ac.kr)
Sangkyu Lee (pharm239@gmail.com)
Phil Hyun Song (sph04@hanmail.net)
Tae Gyun Kwon (tgkwon@knu.ac.kr)
Yun-Sok Ha (yunsokha@gmail.com)

Version: 2 Date: 30 Jul 2020

Author’s response to reviews:

List of responses to the comments of reviewers
We appreciate reviewers the detailed and kind comments for this manuscript. We revised the followings as reviewers’ recommendation. The page numbers below are from the revised manuscript.

[Reviewer 1]
1. In the revised version, BURO-D-20-00279_R1, most concerns have been revised.

But it is still disturbing for the Minor comments point 5:
In Figure 3a, the wound healing assay of T24, it seems that cells in 24h were not seen. Was it the resolution of the figure? It should be revised. It was acceptable in Figure 3b.
The authors response the replacement of the figure with better pictures. In the revised manuscript, the authors include two Figure 3, the first one put before Figure 1. The second Figure 3 was the same as Figure 4. The replaced Figure 3a is still inconvincible.

Response: We apologize for the mistake in uploading revised figures. We replaced Fig 3A to more clear pictures of T24 cells. Additionally we added digital image proof.